Author's response to reviews

Title: Study protocol: The Adherence and Intensification of Medications (AIM) study - a cluster randomized controlled effectiveness study

Authors:

Michele Heisler (mheisler@umich.edu)
Timothy P Hofer (thofer@umich.edu)
Mandi L Klamerus (mandi.klamerus@va.gov)
Julie Schmittdiel (Julie.A.Schmittdiel@kp.org)
Joe Selby (Joe.Selby@kp.org)
Mary M Hogan (mary.hogan@va.gov)
Hayden B Bosworth (boswo001@mc.duke.edu)
Adam Tremblay (astrembl@umich.edu)
Eve A Kerr (ekerr@umich.edu)

Version: 3 Date: 30 August 2010

Author's response to reviews:

Dear Editors,

Per your request, we have removed the conclusion subheading in our manuscript. No additional changes were made to the manuscript.

Sincerely,

Michele Heisler, MD, MPA